

Blood Pressure

BLOOD Pressure



The inverse and non-linear association between central augmentation index and heart rate variability in a cohort of male British combat personnel- findings from the ADVANCE study

Rabeea Maqsood, Susie Schofield, Alexander N Bennett, Ahmed Khattab, Anthony M J Bull, Nicola T Fear, Paul Cullinan, Christopher J Boos & for the ADVANCE Study

To cite this article: Rabeea Maqsood, Susie Schofield, Alexander N Bennett, Ahmed Khattab, Anthony M J Bull, Nicola T Fear, Paul Cullinan, Christopher J Boos & for the ADVANCE Study (23 Jun 2025): The inverse and non-linear association between central augmentation index and heart rate variability in a cohort of male British combat personnel- findings from the ADVANCE study, Blood Pressure, DOI: <u>10.1080/08037051.2025.2524409</u>

To link to this article: https://doi.org/10.1080/08037051.2025.2524409

9	© 2025 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group	+	View supplementary material 🖸
	Accepted author version posted online: 23 Jun 2025.		Submit your article to this journal 🖸
lıl.	Article views: 70	Q	View related articles 🖸
CrossMark	View Crossmark data 🗗		

Check for upda	ates
----------------	------

The inverse and non-linear association between central augmentation index and heart rate variability in a cohort of male British combat personnelfindings from the ADVANCE study

Rabeea Maqsood^{a,*} 0000-0002-8551-0442, Susie Schofield^b, Alexander N Bennett^{a,c}, Ahmed Khattab^d, Anthony M J Bull^a, Nicola T Fear^e, Paul Cullinan^f and Christopher J Boos^{d,g} for the ADVANCE Study

^aCentre for Injury Studies, Department of Bioengineering, Imperial College London, W12 0BZ

^bNational Heart and Lung Institute, Faculty of Medicine, Imperial College London, SW3 6LR ^cAcademic Department of Military Rehabilitation, Defence Medical Rehabilitation Centre, Stanford Hall Estate, Near Loughborough, Nottinghamshire, LE12 5QW ^dDepartment of Medical Sciences and Public Health, Faculty of Health & Social Sciences, Bournemouth University, Bournemouth, BH8 8GP

^eAcademic Department of Military Mental Health and King's Centre for Military Health Research, King's College London, SE5 9RJ

^fLondon, United Kingdom.

^gDepartment of Cardiology, University Hospitals Dorset, Poole Hospital, Poole, BH15 2JB

Corresponding author: Rabeea Maqsood, Centre for Injury Studies, Department of Bioengineering, Imperial College London, W12 0BZ. Email: r.maqsood21@imperial.ac.uk

Abstract

Purpose: The central augmentation index (cAIx) is an indirect measure of arterial stiffness. The influence of heart rate variability (HRV) on cAIx remains unexplored in a military cohort and was the aim of this analysis.

Method: The first follow-up data from the ArmeD serVices trAuma rehabilitatioN outComE (ADVANCE) study were analysed. Participants were male British servicemen who served in Afghanistan (2003-2014) and were divided into two groups at recruitment: injured (who sustained severe combat injury) and uninjured. The uninjured were frequency-matched to the injured by age, rank, role-in-theatre and deployment. HRV was reported as root-mean-square-of-successive-differences (RMSSD) using a five-minute single-lead electrocardiogram. The cAIx was measured using pulse waveform analysis and was adjusted for heart rate at 60 beats/minute (cAIx@60). Effect modification by injury was assessed via interaction analysis. Linear models reported the association between RMSSD (HRV) and cAIx@60 adjusting for a priori confounders.

Results: 1052 participants (injured n=526; uninjured 526; median age at follow-up 37.4 years) were examined. Effect modification by injury was not statistically significant; therefore, was adjusted for along with other confounders. RMSSD and cAIx@60 exhibited a moderate inverse correlation (-0.40; p<0.001). The association between natural log-transformed RMSSD (LnRMSSD) and cAIx@60 was non-linear and statistically significant, suggesting that a 10% decrease in LnRMSSD would be associated with 0.30% increase in cAIx@60.

Conclusion: Lower RMSSD (HRV) is associated with an increase in cAIx@60, independent of injury status and other traditional cardiovascular risk factors. The efficacy of positive HRV modification on cardiovascular risk in military populations needs to be examined.

Plain Language Summary

The hardening of arteries can lead to serious cardiovascular sequelae and adversely affect health and wellbeing of an individual. One way to measure hardening of arteries (also known as arterial stiffness) is through the use of central augmentation index (cAIx). The cAIx measures the extra 'workload' the heart faces due to 'augmented pressure' or the 'extra pressure' exerted on the heart due to hardening of arteries. Previous research has not investigated whether there is a relationship between cAIx and heart rate variability (a measure of autonomic function), especially in a contemporary military cohort. This is crucial as servicemen may have an increased cardiovascular disease risk due to combat exposure and combat injuries, and studying this association may have important implications for their recovery. Therefore, the goal of this analysis was to explore the association between cAIx and HRV. With multivariable regression analysis and after controlling for confounders, we found an inverse and non-linear association between cAIx and HRV in a sample of British military cohort. It suggests that an increase in cAIx leads to a decrease in HRV but not in a linear way. Leveraging on this association, the results suggest the need to explore HRV elevating activities to mitigate the risk of arterial stiffness in military populations.

A Certe

3

Graphical Abstract

The inverse and non-linear association between central augmentation index and heart rate variability in a cohort of male British combat personnel- findings from the ADVANCE study



¹Centre for Injury Studies, Department of Bioengineering, Imperial College London, W12 082. ²National Heart and Lung Institute, Faculty of Medicine, Imperial College London, SW3 6LP. ³Academic Department of Military Rehabilitation, Defence Medical Rehabilitation Centre, Stanford Hall Estate, Neur Loughborough, Nottinghamshire, LE12 50W ⁴Department of Military Rehabilitation Centre, Stanford Hall Estate, Neur Loughborough, Nottinghamshire, LE12 50W ⁴Department of Medical Sciences and Public Health, Faculty of Health & Social Sciences, Bournemouth University, Bournemouth, BH8 BGP. ³Academic Depart of Military Mental Health and King's Centre for Military Health Research, King's College London, SE5 9RJ. ⁹London, United Kingdom. ⁷Department of Cardiology, University Hospitals Dorset, Poole Heapital, Poole, BH15 2JB

Keywords: Arterial stiffness, HRV, vagal tone. Military, Trauma

Accepted

4

Introduction

Heart rate variability (HRV) is a non-invasive and objective measure of cardiac autonomic regulation, indicating balance of underlying sympathetic and parasympathetic activity [1, 2]. HRV is a recognised cardiovascular risk marker; lower HRV is associated with a broad range of cardiovascular risk factors such as type-2 diabetes mellitus [3], obesity [4], and hypertension [5] in healthy and patient populations. Furthermore, among patients with established cardiovascular disease, lower HRV is associated with a worse prognosis [6].

It has also been shown that HRV is inversely associated with arterial stiffness [7]. However, most of the previous research relates to pulse wave velocity (PWV) as a measurement of arterial stiffness. The central arterial augmentation index (cAIx) is another indirect measure of arterial stiffness that reflects the additive effect of reflected arterial pressure waves, at sites of flow impedance, upon central arterial systolic pressure [8]. Stiffer arteries lead to earlier and greater augmentation pressure and cAIx [9]. Higher cAIx leads to increased left ventricular afterload and eventually left ventricular hypertrophy and diastolic dysfunction [10].

The cAIx is under the influence of autonomic nervous system control; increased sympathetic activation leads to arterial vasoconstriction and would be expected to increase central augmentation pressure and cAIx [11]. Unfortunately, the association between HRV and cAIx and the influence of autonomic changes on this relationship have been barely explored [12, 13, 14].

There is evidence to suggest that combat-related traumatic injury (CRTI) is associated with increased cardiovascular risk [15, 16]. Baseline data from the ArmeD serVices trAuma rehabilitatioN outComE (ADVANCE) study has shown significantly lower HRV [17] and higher cAIx in individuals with CRTI versus their uninjured counterparts of similar age, sex (all men), rank and deployment period (Afghanistan) [15]. However, the influence of CRTI on the relationship between HRV and arterial stiffness has not been examined. This is an essential step to determine the potential causal relationship between autonomic (dys)function and increased cAIx.

This study aimed to examine the association between HRV and cAIx (as a measure of arterial stiffness) independent of other cardiovascular risk factors in a cohort of injured and frequency-matched uninjured servicemen. We hypothesised that lower HRV would be

associated with higher cAIx, independent of the injury status and traditional cardiovascular risk factors.

Methods

Study design and setting

The ADVANCE study is a prospective longitudinal cohort study. It aims to investigate the long-term psychological and physiological health impacts of CRTI in a cohort of British military combat veterans and personnel over the course of 20 years [18]. The ADVANCE study completed baseline data collection in August 2020. We undertook the analysis of the first follow-up data of the ADVANCE study. A total of 1053 participants attended the first follow-up assessment (lost to follow-up/deceased/declined follow-up, n=92) as compared to the baseline assessment (n=1145). One participant from the uninjured group sustained a non-military related injury and was excluded in the present analysis to avoid bias, leaving the total sample size of 1052.

The ADVANCE study has full ethics approval from the UK Ministry of Defence Research Ethics Committee (MoDREC) (protocol no:357/PPE/12). All participants voluntarily took part in the study and provided their informed and written consent on the day of the data collection. The data used in the present study was collected between October 2019 and August 2023 by trained research nurses at the Defence Medical and Rehabilitation Centre, Stanford Hall, Loughborough, UK. The full protocol of the ADVANCE study can be found elsewhere [18].

Study population

Participants were male servicemen who had served in the UK Armed Forces during Operation HERRICK in Afghanistan (2003-2014). The participants were divided into two groups: injured (those who sustained a serious CRTI and required aeromedical evacuation to a UK hospital) and uninjured. The uninjured participants were frequency-matched to the injured participants based on age, rank, role-in-theatre, and deployment period [18]. None of the participants had an established cardiovascular, renal, or liver disease or diabetes mellitus prior to the injury/deployment. However, as the present analysis reports on the first follow-up data, the effect of chronotropic medication (beta-blockers) on the reported association between cAIx and HRV was assessed by excluding those participants who consumed beta-blockers in the sensitivity analysis to report unbiased results.

Data collection

Participants visited Stanford Hall, Loughborough for a single study visit which lasted for 6 hours and 30 minutes, on average, including the periods of rest and break between assessments. Participants were requested to refrain from caffeine, alcohol, and food intake for at least eight hours and from smoking for at least four hours prior to data collection. After seeking participants informed written consent, fasting venous blood samples were taken and processed by a local hospital laboratory. These blood samples were used to measure glycated haemoglobin [(HbA1c) as a measure of average glucose control over the preceding three months], high-density lipoprotein cholesterol (HDL-C), low-density lipoprotein cholesterol (LDL-C), and triglycerides.

Following this, pulse wave data for pulse wave analysis (PWA) and pulse wave velocity (PWV) were collected using the VicorderTM device (Skidmore Medical Limited, Bristol, UK) with participants at rest in the supine position. Brachial arterial systolic and diastolic blood pressures, and central arterial augmentation index (cAIx) were measured using the VicorderTM device during the PWA in the morning. The primary outcome of the present analysis was cAIx which was calculated using the average of three cAIx measurements. However, a measurement differing by more than two-fold from the other two measurements in sequence, was treated as an outlier and excluded. Each reading was adjusted for a heart rate of 60 beats/minute (denoted as cAIx@60, %) as previously validated and described [15, 19].

This was immediately followed by HRV assessment. HRV was measured using the Bittium FarosTM ECG device (Mega Motion Faros 180 recorder; Mega, Finland) with participants in the supine position at rest on a hospital bed in a noise free and temperature-controlled room. A continuous single-lead electrocardiogram (ECG) was recorded for 15-20 minutes. After the initial few minutes of setup adjustment and familiarisation, data were collected under spontaneous breathing for five minutes, which was then immediately followed by a slow-paced breathing protocol for five minutes. The primary independent variable of interest in the present analysis was the root mean square of successive differences (RMSSD), a time-domain measure of HRV that represents parasympathetic or vagal tone [1, 20] obtained via above-described five-minute ECG recording. The five-minute RMSSD is a gold-standard measure of short-term HRV and has been considered the best measure for autonomic nervous system activity [21]; it is less susceptible to respiratory changes and has better statistical properties which allow robust calculation [20]. In the present analysis, the five-minute HRV

data recorded under spontaneous breathing protocol was used and reported based on our previous work [22].

Data on socio-demographic, lifestyle (including smoking status), medication history, mental and physical health were collected from participants' clinical interview with a trained research nurse and validated questionnaires as described in the ADVANCE study protocol [18].

In the second half of the study day, data from Dual-Emission X-ray Absorptiometry (DEXA), spirometry, 6-mninute walk test and audiometry were collected. In this analysis, the sixminute walk distance test (6MWD-T) and total Metabolic Equivalents (METs) from all activity sources derived from the International Physical Activity Questionnaire (IPAQ) were used to reflect physical function and physical activity, respectively. The study day ended with a participant debrief with a member of ADVANCE research staff [18].

Injury severity was assessed using the New Injury Severity Scores (NISS) (range 1-75); which were calculated using the Abbreviated Injury Score (2008 updates provided by the UK joint Trauma Theatre Registry; JTTR). The NISS were categorised into higher injury severity (NISS>25 and lower injury severity (NISS<25) based on the previous evidence supporting its link to all-cause mortality [23].

The injury mechanism was categorised into blast versus gunshot wounds and others. Rank at sampling was used as a proxy measure of socioeconomic status and was classified into junior (junior non-commissioned officers and other lower ranks, OR2-4), middle (senior non-commissioned officers, OR5-9), and senior (commissioned officers, OF1-6) [24] ranks, as previously used [15]. Ethnicity was categorised into two groups: White and other ethnic groups.

Study data were collected and managed using Research Electronic Data Capture (REDCap) tool hosted at Imperial College London. REDCap is a secure, web-based software platform designed to support data capture for research studies.

HRV data analysis

The ECG recordings were analysed using Kubios Premium (Kubios Premium V.3.2, The Biomedical Signals Analysis and Medical Imaging Group, University of Kuopio, Finland) for HRV measurement [17, 25]. ECG data were visually inspected for ectopic beats and noise and were manually corrected wherever applicable. This was also supplemented with noise set

to medium and automatic correction. Throughout the analysis, the following operators were kept constant: the smoothness priors (500), interpolation (cubic spline: 4 Hz with 50 ms R-R threshold), and the noise level (medium).

Statistical analysis

Results are presented as mean and standard deviation (SD) for normally distributed data or median and interquartile range (IQR) for skewed data. All data were visually inspected for normality using histograms and QQ plots.

The association between RMSSD and cAIx@60 was assessed using linear models and was tested for linearity using the 'boxtid' command where p < 0.05 was considered to indicate a non-linear association. A statistically significant non-linear relationship between RMSSD and cAIx@60 was observed. Both splines and log-transformation were used to model the nonlinear relationship between RMSSD and cAIx@60. As log-transformation produced lower Bayesian Information Criterion (BIC) than spline, RMSSD was naturally log-transformed (hereby referred to as LnRMSSD) and reported in the the multivariable regression analysis. All models were adjusted for a priori confounders (injury status, age, rank, ethnicity, HDL-C, LDL-C, systolic blood pressure, diastolic blood pressure, triglycerides, and smoking), based on previous literature [13, 14]. The regression coefficient for the association between LnRMSSD and cAIx@60 was interpreted as for each 10% decrease in RMSSD the outcome (cAIx@60) increased by x units calculated by coefficient multiplied by log(1.10), as described [26]. Additional models were run adjusting for injury mechanism, amputation status and injury severity. The relationship between cAIx@60 and RMSSD was visualized using the 'f able' and 'margins' commands in Stata. The 'vif' command in Stata was used to rule out any multi-collinearity.

CRTI was assessed as an effect modifier of the association between LnRMSSD and cAIx@60. An interaction term between CRTI and LnRMSSD was considered [27] adjusting for a priori confounders. In further exploratory analyses, the role of injury mechanism, amputation and injury severity as effect modifiers was also examined via interaction. A sensitivity analysis was performed to exclude those on beta-blockers (n=11) in the model adjusted for injury status and other confounders

RMSSD had 78 (7.4%) missing data points (HRV testing not implemented at the time of assessment n=66; technical issues in ECG recordings n=11; an uninjured participant excluded due to a non-military injury n=1) and there were four missing values (0.4%) for cAIx@60.

The other missing data points were systolic blood pressure (n=1; 0.09%), diastolic blood pressure (n=1; 0.09%), HDL-C (n=23; 2.18%), LDL-C (n=47; 4.46%), triglycerides (n=21; 1.99%), HbA1c (n=24; 2.28%), and smoking (n=26; 2.47%). Multiple imputations were performed using the multiple imputations by chained equations (MICE) algorithm under the assumption that the data were missing at random (MAR) [28] with regress (cAIx@60, HDL-C, LDL-C, LnRMSSD), predictive mean matching (triglycerides, HbAIc, systolic and diastolic blood pressures) and logit methods (smoking) depending on the variables [28, 29]. Auxiliary variables (mean heart rate from the pulse wave velocity measurement and body mass index adjusted for amputation) were added to the imputed model due to their high correlation with the missing data (RMSSD) in the model. These auxiliary variables plus all the variables included in the final model were used in the imputed model. The fit of the imputed model was visually inspected using the 'midiagplot' command in Stata.

The number of imputed datasets (m=50) was decided following the fraction of missing information in the post-model estimates. Complete case and multiple imputed models are reported. The significance level was set at p<0.05. All statistical analyses were conducted in Stata V.17.0 (StataCorp, College Station, Texas, USA).

Results

Sample characteristics

In total, 1052 participants were included; 526 were injured [including 140 amputees] and 526 were uninjured. The median age of participants at the first follow-up assessment was 37.4 years (IQR: 34.15, 41.20). Most of the participants were White and held a junior rank at injury/deployment. Blast was the most common mechanism of injury (71% of the injured). The median NISS was 12 (IQR 5, 22). The injured group had a significantly lower 6MWD and had a significantly higher abdominal waist circumference than the uninjured group (Table 1).

Effect modification by injury status

The effect modification by CRTI on the association between LnRMSSD and cAIx@60 was non-signficant [coefficent -0.23 (95% CI -1.78, 1.31); p =0.76]. The same non-signficant effect modification was observed for higher injury severity [coefficent 0.91 (95% CI -1.73, 3.57), p =0.49], amputation [coefficent 0.61 (95% CI -1.76, 3.00); p =0.61], and injury mechanism (blast) [coefficent 0.13 (95% CI -1.54, 1.82); p =0.87].

Association between RMSSD and cAIx@60

There were no significant differences in systolic and diastolic blood pressures, lipid profile (LDL-C, triglyceride), cAIx (adjusted and unadjusted), and average blood glucose levels (HbA1c) between the injured and the uninjured groups. HDL-C was significantly lower in the injured than in the uninjured (p=0.01). The injured group had a significantly higher resting mean heart rate and lower RMSSD than the uninjured group (Table 2). For the total sample, the cAIx@60 had a significant and negative correlation with RMSSD [Spearman's rho: -0.40 (-0.45, -0.35); p<0.001] (Table 3).

After adjusting for a priori confounders, a10% decrease in RMSSD was associated with a 0.30 increase in cAIx@60 (%) (Table 4). Similar results were observed when excluding those taking beta-blockers (n=11) and with the imputed model (Table 4). There was an inverse and non-linear relationship between cAIx@60 and RMSSD (Figure 1). The additional models adjusted for injury severity, amputation, and injury mechanism also showed a similar coefficient (Table 4).

certer i

Discussion

The principle finding of this analysis is that lower short-term RMSSD (HRV) is inversely correlated and significantly associated with higher cAIx@60 (arterial stiffness, a cardiovascular risk factor). The relationship between RMSSD and cAIx@60 was independent of the injury status and other traditional cardiovascular risk factors. Our findings indicate that a 10% decrease in the RMSSD would be associated with a 0.30% increase in cAIx@60.

The inverse relationship between RMSSD (HRV) and cAIx@60, suggesting that lower HRV is associated with higher arterial stiffness, is not surprising. While the evidence on the topic is scarce, some general trends are comparable. For example, a significant and inverse association between the standard deviation of normal-normal intervals (SDNN) and AIx@75 has been reported in type-1 Diabetes patients [13]. However, our findings are contradictory to the findings of another study [12]. The authors reported a significant positive association between high frequency (LnHF) and AIx@75 in a much smaller study in men (n=39) but not women (n=41) during paced breathing at 12 breaths/minute. Whereas, in another study (n=107) this same group observed that higher heart rate recovery (suggestive of increased cardiac vagal activity) was associated with lower AIx@75 [30]. However, the direct comparison of our findings with other studies is limited given the heterogeneity in HRV measures reported (RMSSD, SDNN, and HF), population type (healthy, diabetes patients and servicemen with and without injury), and adjustment of AIx (at heart rate).

From a physiological standpoint, the precise mechanism of how lower RMSSD (HRV) is associated with higher cAIx@60 is unclear. From the lens of autonomic imbalance, lower RMSSD reflects vagal withdrawal [20], which may result in elevated sympathetic activity [31]. The positive correlation between sympathetic activity and AIx [32] may be a plausible explanation for this observation. Furthermore, increased vasoconstriction due to enhanced sympathetic activity affects vascular tone [11, 33] and leads to relative vascular impedance and increased augmentation index. The relationship between HRV and cAIx@60 may be bidirectional, given the suggestion of a bidirectional relationship between HRV and PWV-another marker of arterial stiffness [34]. The possibility of the influence of mediating factors (not included in this analysis) such as depression, anxiety, and physical function on the causal pathway between RMSSD and cAIx@60 cannot be excluded.

The observation that combat injury (and other injury variables) did not significantly moderate the relationship between cAIx@60 and RMSSD was unexpected. Our findings indicate that a

decrease in RMSSD would be associated with an increase cAIx@60 to almost the same degree in the injured as in the uninjured. While in the present analysis, the injury status was included as a confounder rather than a moderator, it will be the subject of our future work to investigate if this is a recurring trend in the upcoming follow-ups of the ADVANCE study.

Another finding of this analysis is the non-linear and inverse relationship between RMSSD and cAIx@60. The cAIx@60 appears to have a steeper decline at the lower end of the scale with narrow CIs and then becomes less steep as RMSSD increases with relatively wider CIs. The explanation of this trend is limited given the paucity of evidence on the link between RMSSD and cAIx@60.

To the authors' knowledge, it is the first study to have provided preliminary evidence on the association between HRV and cAIx@60 independent of traditional cardiovascular risk factors in a contemporary military cohort of injured and uninjured veterans and personnel. Another key strength of this analysis is its large sample size as compared to previous studies with a sample size of <550 [13, 30]. An exploratory analysis was conducted to examine the influence of other aspects of injury (mechanism, severity and amputation status) for a comprehensive assessment of the RMSSD-cAIx@60 association. A standardised protocol of data collection was used for all variables of interest indicating the internal validity of this analysis. Multiple imputation was also performed to account for the missing data and to report unbiased results.

The generalisability of results may be limited to populations similar in characteristics to participants included in this study – adult, mainly White male servicemen. The sensitivity analysis showed similar results when excluding those taking beta-blockers (n=11). Given this, participants taking beta-blockers were included in the reported analysis. Given the negative correlation between heart rate and AIx [19], AIx was adjusted at a standard heart rate of 60; however, the AIx and RMSSD models were not adjusted for heart rate to protect and maintain the intrinsic physiological influence of heart rate on RMSSD (HRV).

It may be early to ascertain the clinical significance of our findings given the exploratory nature of this analysis. Nonetheless, our findings present important implications for practice and research and should be contextualised for interpretation. Our findings suggest that increasing HRV levels in combat veterans might mitigate the risk of early atherosclerosis (cAIx@60) and potentially a later cardiovascular event. While HRV stands out as a robust non-invasive biomarker of cardiovascular health, its link with AIx and other lifestyle factors

remains under-represented in current research. Given this, we call for further research in the field to investigate the link between cAIx@60 and HRV in military cohorts as well as in other vulnerable and under-represented populations to generate comparable evidence. The causal pathway between cAIx and HRV also warrants exploration to fully understand the effect of HRV on cAIx in combat veterans and other populations.

In summary, in a large, combined group of injured and uninjured servicemen, lower RMSSD was associated with cAIx@60 in a non-linear pattern, independent of injury status and traditional cardiovascular risk factors.

nusó certek

Acknowledgments: We wish to thank all the research staff at Stanford Hall who helped with the ADVANCE study, including Emma Coady, Grace Blissitt, Melanie Chesnokov, Daniel Dyball, Sarah Evans, Guy Fraser, Nicola Goodman, Alison Hever, Meliha Kaya-Barge, Jocelyn Keshet-Price, Eleanor Miller, Steven Parkes, Bharti Patel, Samantha Paul, David Pernet, Vlad Pop, Helen Prentice, Ursula Pucilowska, Stefan Sprinckmoller, Jodie Stevenson, Lalji Varsani, Anna Verey, Molly Waldron, Owen Walker, Farheen Dairkee, Tasarla White, Seamus Wilson, and Severija Juškaitė.

Declaration of Interest Statement: ANB is a serving member of the Royal Air Force. NTF is a trustee of Help for Heroes and is part funded by a grant from the UK Ministry of Defence. All the remaining authors declare no conflicts of interest. The views expressed are those of the authors and not necessarily those of the UK Ministry of Defence.

Author contribution: RM and CJB contributed to the study conception and design. AK acquired the funding for RM's PhD studentship at Bournemouth University. Material preparation and data management were performed by RM. HRV analysis was performed by RM and supervised by CJB. Statistical analyses were performed by RM and supervised by SS. The first draft of the manuscript was written by RM. The project was overall supervised by AK and CJB. The manuscript was critically reviewed and suggested improvements related to intellectual content by ANB, SS, NTF, AMJB, PC, and AK. All co-authors read and approved the final version of the manuscript.

Ethical approval statement: This study had full ethics approval from the UK Ministry of Defence Research Ethics Committee (protocol no:357/PPE/12). All participants in this study undertook full informed and written consent. This study was conducted in compliance with the Declaration of Helsinki (1964).

Data availability statement: Given the unusually sensitive nature, data have not been made widely available. Requests for data will be considered on a case-by-case basis and subject to UK Ministry of Defence approval and clearance. The ADVANCE study protocol is available online.

Funding statement: This study is a part of RM's PhD studentship- jointly funded by Bournemouth University and the ADVANCE study, UK. The ADVANCE study (grant number: ADV-ADMR-03) is funded through the ADVANCE Charity. Key contributors to the charity are the Headley Court Charity (principal funder), HM Treasury (LIBOR Grant), Help for Heroes, Nuffield Trust for the Forces of the Crown, Forces in Mind Trust, National Lottery Community Fund, Blesma - The Limbless Veterans, the UK Ministry of Defence, and the Office for Veterans' Affairs (OVA). The funders of the study had no role in study design, data collection, data analysis, data interpretation, or writing of the manuscript.

Received

References

- Malik M, Bigger JT, Camm AJ, Kleiger RE, Malliani A, Moss AJ, Schwartz PJ. Heart rate variability: Standards of measurement, physiological interpretation, and clinical use. Eur. Heart. J. 1996;17(3):354-381.
- 2. Billman GE, Huikuri HV, Sacha J, Trimmel K. An introduction to heart rate variability: methodological considerations and clinical applications. Front. Physiol.2015;6:55.
- Benichou T, Pereira B, Mermillod M, Tauveron I, Pfabigan D, Maqdasy S, Dutheil F. Heart rate variability in type 2 diabetes mellitus: A systematic review and meta-analysis. PLoS. One. 2018;13(4): e0195166.
- 4. Yadav RL, Yadav PK, Yadav LK, Agrawal K, Sah SK, Islam MN. Association between obesity and heart rate variability indices: an intuition toward cardiac autonomic alteration–a risk of CVD. Diabetes. Metab. Syndr. Obes. Targets. Ther. 2017;57-64.
- Hoshi RA, Santos IS, Dantas EM, Andreão RV, Mill JG, Lotufo PA, Bensenor I. Reduced heart-rate variability and increased risk of hypertension—A prospective study of the ELSA-Brasil. J. Hum Hypertens. 2021;35(12):1088-1097.
- 6. Fang SC, Wu YL, Tsai PS. Heart rate variability and risk of all-cause death and cardiovascular events in patients with cardiovascular disease: a meta-analysis of cohort studies. Biol. Res. Nurs. 2020;22(1):45-56.
- Schaarup JR, Christensen MS, Hulman A, Hansen CS, Vistisen D, Tabák AG, Witte DR, Bjerg L. Autonomic dysfunction is associated with the development of arterial stiffness: The Whitehall II cohort. GeroScience. 2023;45(4):2443-2455.
- Laurent S, Cockcroft J, Van Bortel L, Boutouyrie P, Giannattasio C, Hayoz D, Pannier B, Vlachopoulos C, Wilkinson I, Struijker-Boudier H. Expert consensus document on arterial stiffness: methodological issues and clinical applications. Eur. Heart. J. 2006; 27(21):2588-2605.
- 9. Liao J, Farmer J. Arterial stiffness as a risk factor for coronary artery disease. Curr. Atheroscler. Rep. 2014;16:1-7.
- 10. Davies JI, Struthers AD. Beyond blood pressure: Pulse wave analysis-a better way of assessing cardiovascular risk? Future. Cardiol. 2005; 1(1): 69-78.
- 11. Bruno RM, Ghiadoni L, Seravalle G, Dell'Oro R, Taddei S, Grassi G. Sympathetic regulation of vascular function in health and disease. Front. Physiol. 2012;3:284.

- Latchman PL, Thiel R, Gates G, Zhu W, Axtell R, Gardner K, Lunn W, DeMeersman R. Cardiovagal tone: a predictor of heart rate adjusted augmentation index in men but not in women. Open. J. Mol. Integr. Physiol. 2017;7(3):41-51.
- 13. Jaiswal M, Urbina EM, Wadwa RP, Talton JW, D'Agostino Jr RB, Hamman RF, Fingerlin TE, Daniels SR, Marcovina SM, Dolan LM, Dabelea D. Reduced heart rate variability is associated with increased arterial stiffness in youth with type 1 diabetes: the SEARCH CVD study. Diabetes. Care. 2013;36(8):2351-2358.
- 14. TODAY Study Group, A.S Shah, L. El ghormli, S.S Gidding et al., Longitudinal changes in vascular stiffness and heart rate variability among young adults with youth-onset type 2 diabetes: results from the follow-up observational treatment options for type 2 diabetes in adolescents and youth (TODAY) study. Acta. Diabetol. 2022;1-9.
- 15. Boos CJ, Schofield S, Cullinan P, Dyball D, Fear NT, Bull AM, Pernet D, Bennett AN. Association between combat-related traumatic injury and cardiovascular risk. Heart. 2021;108(5):367-374.
- Boos CJ, De Villiers N, Dyball D, McConnell A, Bennett AN. The Relationship between Military Combat and Cardiovascular Risk: A Systematic Review and Meta-Analysis. Int. J. Vasc. Med. 2019; 1-14.
- 17. Maqsood R, Schofield S, Bennett AN, Bull AM, Fear NT, Cullinan P, Khattab A, Boos CJ. Relationship between combat-related traumatic injury and ultrashort term heart rate variability in a UK military cohort: findings from the ADVANCE study. BMJ. Mil. Health. 2023.
- 18. Bennett AN, Dyball DM, Boos CJ, Fear NT, Schofield S, Bull AM, Cullinan P. Study protocol for a prospective, longitudinal cohort study investigating the medical and psychosocial outcomes of UK combat casualties from the Afghanistan war: the advance study. BMJ. Open. 2020;10(10): e037850.
- Wilkinson IB, MacCallum H, Flint L, Cockcroft JR, Newby DE, Webb DJ. The influence of heart rate on augmentation index and central arterial pressure in humans. J. Physiol. 2000; 525(Pt 1): 263.
- 20. Laborde S, Mosley E, Thayer JF. Heart rate variability and cardiac vagal tone in psychophysiological research–recommendations for experiment planning, data analysis, and data reporting. Front. Psychol. 2017;8:213.
- 21. Jarczok MN, Weimer K, Braun C, Williams DP, Thayer JF, Gündel HO, Balint EM. Heart rate variability in the prediction of mortality: A systematic review and meta-analysis of healthy and patient populations. Neurosci. Biobehav. Rev. 2022;143:104907.

- 22. Maqsood R, Schofield S, Bennett AN, Khattab A, Bull AM, Fear NT, Boos CJ. Exploratory analysis of spontaneous versus paced breathing on heart rate variability in veterans with combat-related traumatic injury. PM&R. 2024.
- 23. Li H, Ma YF. New injury severity score (NISS) outperforms injury severity score (ISS) in the evaluation of severe blunt trauma patients. Chin. J. Traumatol. 2021;24(05):261-265.
- Yoong SY, Miles D, McKinney PA, Smith IJ, Spencer NJ. A method of assigning socioeconomic status classification to British Armed Forces personnel. J. R. Army. Med. Corps. 1999;145(3):40-142.
- 25. Tarvainen MP, Niskanen JP, Lipponen JA, Ranta-Aho PO, Karjalainen PA. Kubios HRV– heart rate variability analysis software. Comput. Methods. Programs. Biomed. 2014;113(1):210-220.
- 26. UCLA Statistical Consulting Group (2021). FAQ How do I interpret a regression model when some variables are log transformed? Available from: https://stats.oarc.ucla.edu/other/mult-pkg/faq/general/faqhow-do-i-interpret-a-regression-model-when-some-variables-are-log-transformed/ [Accessed 6 December 2024].
- 27. Marlin N, Allotey J. The difference between effect modification and covariate confounding. BJOG. 2021;128:1574.
- 28. Azur MJ, Stuart EA, Frangakis C, Leaf PJ. Multiple imputation by chained equations: what is it and how does it work?. Int. J. Methods. Psychiatr. Res. 2011; 20(1):40-49.
- 29. Lee KJ, Roberts G, Doyle LW, Anderson PJ, Carlin JB. Multiple imputation for missing data in a longitudinal cohort study: a tutorial based on a detailed case study involving imputation of missing outcome data. Int. J. Soc. Res. Methodol. 2016;19(5):575-591.
- 30. Latchman PL, Yang Q, Kong L, Zhang H, Sebagisha J, De Meersman RE. Heart Rate Recovery, Central Systolic Pressure, and Augmentation Index in Young Healthy Individuals. Vasc. Health. Risk. Manag. 2022;17-25.
- Shaffer F, Ginsberg JP. An overview of heart rate variability metrics and norms. Front. Pub. Health. 2017;5:258.
- 32. Casey DP, Curry TB, Joyner MJ, Charkoudian N, Hart EC. Relationship between muscle sympathetic nerve activity and aortic wave reflection characteristics in young men and women. Hypertens. 2011; 57(3):421-427.
- 33. Amiya E, Watanabe M, Komuro I. The relationship between vascular function and the autonomic nervous system. Ann. Vasc. Dis. 2014; 7(2):109-119.

34. Pilz N, Heinz V, Ax T, Fesseler L, Patzak A, Bothe TL. Pulse Wave Velocity: Methodology, Clinical Applications, and Interplay with Heart Rate Variability. Rev. Cardiovasc. Med. 2024;25(7):266.

Accepted Manuschi

	Total	Injured	Uninjured
Number	1052	526	526
Age at FU1 assessment, years	37.44	37.17	37.86
	(34.15, 41.20)	(34.09, 41.18)	(34.22, 41.21)
Amputee, n (%)			
-Yes	-	140 (26.6%)	
-No	-	386 (73.3%)	-
Injury mechanism, n (%)			
-Blast	-	375 (71.29%)	-
-GSW and others	-	151 (28.71%)	
NISS-2008, median (IQR)		12 (5, 22)	
Injury severity (NISS-2008), n (%)			
-Lower (NISS<25)	-	412 (78.33%)	
-Higher (NISS>25)	-	114 (21.67%)	
Ethnicity, n (%))
-White	957 (90.97%)	479 (91.06%)	478 (90.87%)
-Other ethnic groups	95 (9.03%)	47 (8.94%)	48 (9.13%)
Rank at sampling [†] , n (%)			
-Junior (OR2-4)	677 (64.35%)	368 (69.96%)	309 (58.75%)
-Middle (OR5-9)	242 (23.00%)	100 (19.01%)	142 (27.00%)
-Senior (OF1-6)	133 (12.64%)	58 (11.03%)	75 (14.26%)
Smoking status‡, n (%)		0	
-Never	481 (46.88%)	244 (48.13%)	237 (45.66%)
-Ex-smoker	380 (37.04%)	195 (38.46%)	185 (35.65%)
-Smoker	165 (16.08%)	68 (13.41%)	97 (18.69%)
Height, cm	179.21+6.59	179.46+6.91	178.95+6.25
Weight, kg	89.63+14.23	88.97+15.35	90.30+12.99
BMI adjusted for amputation, kg/m ²	28.33+3.87	28.48+4.07	28.17+3.66
Abdominal waist circumference [‡] , cm	97.09+10.61	97.82+11.26	96.36+9.88
Total weekly activity from all sources [‡] ,	5850	5868	5811
METs	(2784, 10776)	(2340, 10836)	(2977, 10737)
6MWD-T‡, meters	625.69+104.71	597.15+114.95	652.59+85.85

Table 1: Demographic and lifestyle characteristics of the participants by injury status.

-Data presented as mean±SD or number (%) or median (inter-quartile range; IQR) for highly skewed data.

*Based on the comparison between the injured and uninjured participants using appropriate equality test based on normality.

-FU1, first follow-up assessment; GSW, gunshot wound; NISS, New Injury Severity Score; BMI, Body Mass Index; 6MWD-T, 6-minute walk distance test; METs, Metabolic Equivalents.

†Junior non-commissioned officers and other lower ranks (OR2-4), senior non--commissioned officers (OR5-9), commissioned officers (OF1-6)

\$Missing data (6MWDT n= 38; all activities METs n=54; abdominal circumference n=9, smoking status n=26)

	Injured	Uninjured
Number	526	526
Systolic blood pressure†, mmHg	134.13+11.24	134.07+10.95
Diastolic blood pressure†, mmHg	70.34+9.17	69.69+8.87
cAIx† (unadjusted), %	17.63+7.16	17.69+7.50
cAIx [†] (adjusted at heart rate 60), %	17.35+8.34	16.51+8.70
Mean heart rate†, BPM	59.38+9.85	57.18+9.65
RMSSD†, ms	37.71 (25.37, 55.93)	42.07 (27.74, 62.44)
HDL-C†, mmol/L	1.29+0.31	1.34+0.29
LDL-C [†] , mmol/L	3.16+0.85	3.23+0.80
Triglycerides†, mmol/L	1.12 (0.8, 1.6)	1.13 (0.8, 1.61)
HbA1c [†] , mmol/mol	34.61+6.44	34.67+5.16

Haemodynamic, HRV, and cAIx data of the participants by injury status. Table 2:

-Data presented as mean±SD or median (inter-quartile range; IQR) for highly skewed data. *Based on the comparison between injured and uninjured participants using appropriate equality test based on normality. -cAIx, central Augmentation Index; RMSSD, Root Mean Square of Successive Difference; HDL-C, High-Density Lipoprotein Cholesterol;

LDL-C, Low-Density Lipoprotein Cholesterol, HbAIc; Glycated haemoglobin.

†Missing data (systolic/diastolic blood pressures n=1; cAIx (unadjusted) n=1; cAIx@60 (adjusted at heart rate of 60/minute) n=4; HDL-C

n=23; Triglycerides n=21; LDL-C n=47; HbA1c n=24; Heart Rate, n=32; RMSSD n=78).

Correlation between cAIx@60 and RMSSD. Table 3:

×C	Correlation between RMSSD and cAIx@60 (95% CI and <i>p</i> -value)
Total sample	-0.40 (-0.45, -0.35)
	<i>p</i> <0.001
Injured	-0.42 (-0.50, -0.35)
	<i>p</i> <0.001
Uninjured	-0.37(-0.45, -0.30)
N N	<i>p</i> <0.001

Table 4:	Multivariable regression analysis of the cAIx@60 with RMSSD.

	Complete case analysis (n=926)										
	Univariable model	Model 1 (Injury status) [†]		Model 2 (Injury status [†]) excluding beta-blocker‡ users (n=915).		Model 3 (Injury Severity [†])		Model 4 (Amputation [†])		Model 5 (Injury Mechanism [†])	
	Unadjusted unstandardised coefficient. (95% CI)	Adjusted unstandardised coefficient. (95% CI)	<i>p</i> -value	Adjusted unstandardised coefficient. (95% CI)	<i>p</i> -value	Adjusted unstandardised coefficient. (95% CI)	<i>p</i> -value	Adjusted unstandardised coefficient. (95% CI)	<i>p</i> -value	Adjusted unstandardised coefficient. (95% CI)	<i>p</i> -value
LnRMSSD*	-5.09 (-5.88, -4.30)	-3.21 (-4.06, -2.36)	<0.001	-3.23 (-4.09, -2.38)	<0.001	-3.22 (-4.06, -2.37)	<0.001	-3.17 (-4.02, -2.32)	<0.001	-3.20 (-4.05, -2.35)	<0.001
	Multiple imputed model (n=1053)										
LnRMSSD*	-5.27 (-6.07, -4.48)	-3.67 (-4.51, -2.82)	<0.001	-3.66 (-4.52, -2.81)	<0.001	-3.67 (-4.51, -2.82)	<0.001	-3.61 (-4.46, -2.77)	<0.001	-3.65 (-4.50, -2.80)	<0.001

*Ln, Natural log of RMSSD used in the regression. Interpreted as: coefficient*log(1.n) where n= point e.g $3.21*\log(1.10) = 0.30$ difference in cAIx@60 in response to 10% decrease in RMSSD in the injury status model. Uninjured were included as reference category in all injury-related models.

†Model further adjusted for age at the first follow-up assessment, ethnicity, rank at sampling, systolic and diastolic blood pressures, HDL-C, LDL-C, triglycerides, HbA1c, and smoking.

‡In the imputed model without beta blocker users, missing data assumed as non- beta blocker user.



Figure 1: An inverse and non-linear association between cAIx@60 and RMSSD (adjusted for injury status and other confounders).

Received with the second

211